Author's response to reviews

Title: Health behaviour modelling for prenatal diagnosis in Australia: A geodemographic framework for health service utilisation and policy development

Authors:

Evelyne E Muggli (evi.muggli@mcri.edu.au)
David McCloskey (davidm@pathfindersolutions.com.au)
Jane L Halliday (jane.halliday@mcri.edu.au)

Version: 6 Date: 30 August 2006

Author's response to reviews: see over
30 August 2006

The Editorial Team

BMC Health Services Research

Health behaviour modelling for prenatal diagnosis in Australia:
A geodemographic framework for health service utilisation and policy development

Muggli, Evelyne E, evi.muggli@mcri.edu.au,
McCloskey, David, davidm@pathfindersolutions.com.au, and
Halliday, Jane L, jane.halliday@mcri.edu.au,

Dear Editors,

Thank you for accepting our manuscript for publication in BMC Health Services Research.

Please find the revised manuscript attached, with the following changes to the formatting as requested:

a) Author list on title page has been amended as per submission system
b) Highlighting has been removed from manuscript text
c) The heading “conclusions” now precedes the last paragraph of the manuscript
d) The Additional file section had now been moved to after the tables.
e) Rows in tables are now intersected by vertical lines instead of delineated horizontally
f) Table 1a and b are merged into one single table.
g) The keys to Figures 1-3 have been added to each figure image.

We hope that by making these changes we have been able to assist with the online production process and look forward to hearing from you soon.

Sincerely,

Evelyne Muggli
Public Health Genetics